Author's response to reviews

Title: Protective Role of 1,25(OH)2Vitamin D3 in the Mucosal Injury and Epithelial Barrier Disruption in DSS-induced Acute Colitis in Mice

Authors:

hongwei zhao (zhaohongwei788@163.com)
hong zhang (zhanghongykb@126.com)
hui wu (huinewmail@163.com)
hui li (lihuilih@163.com)
lei liu (liuleiketty@yahoo.cn)
jian guo (xinyunjianke@163.com)
chenyang li (810601725@qq.com)
David Q shih (David.shih@csmc.edu)
xiaolan zhang (xiaolanzh@126.com)

Version: 2 Date: 3 April 2012

Author's response to reviews: see over
Dear editor and reviewers,

Manuscript 8280605996726508 entitled "Protective Role of 1,25(OH)2Vitamin D3 in the Mucosal Injury and Epithelial Barrier Disruption in DSS-induced Acute Colitis in Mice" by Hongwei Zhao et al.

We would like to thank the Reviewers and Editorial Board for their valuable time and effort in reviewing our work. Below are our point-by-point response that we have made in the revised manuscript.

All the authors have read the revised manuscript and approved it. This manuscript has not been submitted to or published in any other journal. There are no conflicts declared.

Sincerely yours,
Xiaolan Zhang, M.D., PhD.
Dept of Gastroenterology, the Second Hospital of Hebei Medical University.
No.215 Heping west Road, Shijiazhuang City, Hebei Province, China. 050000
Tel: 13703313189
E-mail: xiaolanzh@126.com
MS: 8280605996726508
Protective Role of 1,25(OH)₂Vitamin D3 in the Mucosal Injury and Epithelial Barrier Disruption in DSS-induced Acute Colitis in Mice Hongwei Zhao¹*, Hong Zhang¹*, Hui Wu¹*, Hui Li¹, Lei Liu¹, Jian Guo¹, Chenyang Li², David Q Shih³, Xiaolan Zhang¹

1. Please include a 'Competing interests' section between the Conclusions and Authors' contributions.
   Answer: Competing interests have been included between the Conclusions and Authors' contributions in the revised manuscript.

2. In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.
   Answer: Now this work was supported by public health bureau grant of Hebei province (No. 20090104) to XL Zhang.

3. Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.
   Answer: we don’t hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future.

4. Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.
   Answer: we are not currently applying for any patents relating to the content of the manuscript. We haven’t received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript.

5. Do you have any other financial competing interests? If so, please specify.
   Non-financial competing interests: are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.
   Answer: we haven’t any other financial competing interests. Non-financial competing interests: there aren’t any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript.